Author’s response to reviews

Title: Cognitive medicine - a new approach in health care science

Authors:
Anders Wallin (anders.wallin@neuro.gu.se)
Per Johansson (pmj@gu.se)
Ingibjörg Jonsdottir Jonsdottir (ingibjorg.jonsdottir@vgregion.se)
Petronella Kettunen (Petronella.Kettunen@neuro.gu.se)
Christer Nilsson (christer.nilsson@med.lu.se)
Michael Nilsson (Michael.Nilsson@hmri.org.au)
Arto Nordlund (anrto.nordlund@neuro.gu.se)
Lars Nyberg (lars.nyberg@umu.se)
Katharina Sunnerhagen (Katharina.Sunnerhagen@neuro.gu.se)
Johan Svensson (johan.svensson@medic.gu.se)
Beata Terzis (Beata.Terzis@frosunda.se)
Lars-Olof Wahlund (Lars-Olof.Wahlund@ki.se)
Georg Kuhn (georg.kuhn@neuro.gu.se)

Version: 4 Date: 18 Jan 2018

Author’s response to reviews:
Re email Matthew Hickey Jan 17, 2018

Many thanks for your letter.

Arto Nordlund is now in the author's list.

The text of Acknowledgements is:
"Arto Nordlund was deceased November 3, 2017. We commemorate him for his important scientific and clinical work not the least his excellent lectures that attracted young people to the field of cognitive medicine."
His initials are in Author's contributions.

For obvious reasons his email address is lacking in the manuscript but it was inserted in the system to enable submission of this revision.